Author's response to reviews

Title: The functional status and well being of people with myalgic encephalomyelitis/chronic fatigue syndrome and their carers

Authors:

Luis C Nacul (luis.nacul@lshtm.ac.uk)
Eliana M Lacerda (eliana.lacerda@lshtm.ac.uk)
Peter Campion (p.d.campion@hull.ac.uk)
Derek Pheby (derekpheby@btinternet.com)
Maria de L Drachler (jcdc.leite@googlemail.com)
Jose C Leite (jcdc.leite@googlemail.com)
Fiona Poland (F.Poland@uea.ac.uk)
Amanda Howe (Amanda.Howe@uea.ac.uk)
Shagufta Fayyaz (Shagufta.Fayyaz@lshtm.ac.uk)
Mariam Molokhia (Mariam.Molokhia@kcl.ac.uk)

Version: 6 Date: 17 May 2011

Author's response to reviews:

Dear Editors,

As requested by Keri Davies, we have removed the additional files.

Regards,

Luis Nacul (on behalf of the authors)